Author's response to reviews

Title: Promoting E2F1-mediated apoptosis in oestrogen receptor-alpha-negative breast cancer cells

Authors:

Maria F Montenegro (fermontenegro@um.es)
Maria del Mar Collado-Gonzales (mdmcg1@um.es)
Maria Piedad Fernandez-Perez (mfp18858@um.es)
Manel B Hammouda (ma_nel85@hotmail.com)
Lana Tolordava (lanatolordava@mail.ru)
Mariam Gamkrelidze (gamkrelidzem@yahoo.com)
Jose Neptuno Rodriguez-Lopez (neptuno@um.es)

Version: 2  Date: 16 December 2013

Author's response to reviews: see over
Dear Editor,

Please find attached our manuscript entitled “Promoting E2F1-mediated apoptosis in estrogen receptor-α-negative breast cancer cells” by Montenegro et al.

Now we have included the Editorial Requests in the new version of the manuscript.

1. We have transferred the Authors’ Details and their email addresses to the Title Page.
2. This manuscript does not include research involving human subjects (including human material or human data) nor primary cell lines obtained from patients. We used established human cells lines obtained from the ATCC. Under our understanding, and also confirmed for the Ethical Committee of our Institution, this material does not need approval from an appropriate ethics committee. Please inform us if we are wrong.

José Neptuno Rodríguez-López, on behalf of the authors.